Author’s response to reviews

Title: The medium-term sustainability of organisational innovations in the National Health Service

Authors:

Graham Martin (gpm7@le.ac.uk)
Graeme Currie (graeme.currie@wbs.ac.uk)
Rachel Finn (r.l.finn@sheffield.ac.uk)
Ruth McDonald (ruth.mcdonald@nottingham.ac.uk)

Version: 2 Date: 1 March 2011

Author’s response to reviews:

Dear Sir/Madam

Many thanks for copy-editing our manuscript. We have been through the manuscript and have made changes requested by the copy editor (these are tracked in the newly uploaded version). We have not changed the abstract as requested, as the guidance on the website seems to indicate that the format we have used for the abstract is the correct one for protocol papers, and not the format suggested by the copy editor. Please let us know if this does need to be changed.

Yours faithfully

Ruth McDonald